Author's response to reviews

Title: Water displacement leg volumetry in clinical studies - A discussion of error sources

Authors:

Eberhard Rabe (Eberhard.Rabe@ukb.uni-bonn.de)
Markus Stücker (M.Stuecker@derma.de)
Bertram Ottillinger (volumetry@vicron.de)

Version: 2 Date: 30 April 2009

Author's response to reviews:

Manuscript ID 1074165259251862
Water displacement leg volumetry in clinical studies - A discussion of error sources
Rabe E, Stücker M, Ottillinger B

Dear Editors,

We had recently submitted our paper on water displacement leg volumetry to BMC Cardiovascular Disorders. Following discussions with and suggestions of Dr. Graham, we have decided to re-submit this paper to BMC Medical Research Methodology as a debate. In this course we have, as requested, removed our unpublished data and focused the article on previously published studies.

We will submit the unpublished results as a separate Research Article in due course.

With kind regards

Dr. med. Bertram Ottillinger